Author's response to reviews

Title: The utility of the Historical Clinical Risk -20 Scale as a predictor of outcomes in decisions to transfer patients from high to lower levels of security-A UK perspective.

Authors:

Mairead Dolan (mairead.dolan@forensicare.vic.gov.au)
Regina Blattner (regine.blattner@mhsc.nhs.uk)

Version: 4 Date: 7 September 2010

Author's response to reviews: see over
7th September 2010

Dear Editor,

Re: The utility of the Historical Clinical Risk -20 Scale as a predictor of outcomes in decisions to transfer patients from high to lower levels of security-A UK perspective.

Please find attached the final revised version of the above manuscript for publication in BMC Psychiatry. As suggested in your email of 26th August 2010, we have:

a. Replaced the term "primary personality disorders" from table 2
b. Swapped the limitations and conclusions sections
c. Inserted an acknowledgement section

Yours Sincerely,

M. Dolan

Professor Mairead Dolan